Author's response to reviews

Title: Effects of resuscitation with crystalloid fluids on cardiac function in patients with severe sepsis

Authors:

Zhi Xun Fang (fangzhixun@yahoo.com.cn)
Yu Feng Li (liyufeng99@netease.com)
Xiao Qing Zhou (zxq@gcph.com.cn)
Zhen Zhang (zzlp2000@yahoo.com.cn)
Jin Song Zhang (zhangjso@sina.com)
Hai Ming Xia (xiahaiming700611@163.com)
Guo Ping Xing (wgzx@gcph.com.cn)
Wei Ping Shu (swp2008@yeah.net)
Ling Shen (najingshl@163.com)
Guo Qing Yin (yinguoq@jlonline.com)

Version: 5 Date: 10 April 2008

Author's response to reviews: see over
Dear Editors,

We have revised our manuscript during proof-reading. Some errors in the present manuscript were corrected as follows:

In the second line of “Email” section on 1 page, “Zhen Zhang, of the emergency department of the First Hospital of Nanjing, the conduction of the study and performer of clinical trial; zzlp2000@yahoo.com.cn” was changed into “Zhen Zhang - zzlp2000@yahoo.com.cn”.


In “Figure legends section”:
In Fig 1, the first sentence “The trial program and case grouping were carried out according to the design indicated in Materials and Methods.” was changed into “The trial program and case grouping were carried out according to the design indicated in Methods.”

The titles in the image files have been deleted, as required.

We will pay the process charge by remitting through Bank of China. We will also order 100 copies of offprint and pay for them. We wonder if we could remit the pay together with the process charge and if you could send us an invoice that includes the two sums of money we pay you. Would you please send us the electronic copy of the invoice by email so that we will be able to remit to you the money immediately?

Thank you!

Yours sincerely,

Dr. Guo Qing Yin